Author's response to reviews

Title: Sodium hyaluronate and chondroitin sulfate replenishment therapy can improve nocturia in men with post-radiation cystitis: results of a prospective pilot study

Authors:

Mauro Gacci (maurogacci@yahoo.it)
Omar Saleh (ps11nov@hotmail.com)
Claudia Giannessi (cgiannessi@yahoo.it)
Beatrice Detti (beatrice.detti@aouc.unifi.it)
Lorenzo Livi (lorenzo.livi@unifi.it)
Eleonora Monteleone Pasquetti (flowerle@tiscali.it)
Tatiana Masoni (tatim@tiscali.it)
Enrico Finazzi Agro’ (efinazzi@tin.it)
Vincenzo Li harizi (vlimarzi@hotmail.com)
Andrea Minervini (andreamine@libero.it)
Marco Carini (carini@unifi.it)
Stavros Gravas (sgravas2002@yahoo.com)
Matthias Oelke (Oelke.Matthias@mh-hannover.de)
Sergio Serni (sergio.serni@unifi.it)

Version: 4 Date: 12 March 2015

Author's response to reviews:

Copy-edit

We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/info/authors/authorfaqs#12.

RE: Thank you for the advise. As you suggest we resubmitted the papaer to aour english expert that confirm it acceptable as underlined to the reviewer

2. In the list of author e-mail addresses MG’s e-mail address is mistakenly given as maurogacci@yoahoo.it and needs to be corrected.

RE: Done it. Thank you.

3. Format Author's Contribution Section according to our guidelines.
We suggest the following format (please use initials to refer to each author’s contribution): AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript.

Contributors who do not meet the criteria for authorship should be listed in an acknowledgements section. Examples of those who might be acknowledged include a person who provided purely technical help, writing assistance, or a department chair who provided only general support.

RE: Done it. Thank you.

4. Include details of any funding in the Acknowledgements.

RE: We received by IBSA Farmaceutici only the drugs as mentioned in the manuscript. No other financial support.

5. Figure legend needs to be moved to after the References.

RE: Done it. Thank you.

6. Format References according to our guidelines

(a) Reference citations: All references must be cited in the text following journal style. Please see the instructions for authors on the journal website for further information and adjust your references accordingly.

RE: Done it. Thank you.

(b) References in additional files: Please be advised that references cannot be cited only in the additional file section of the manuscript text; please add citations to these references into the main text instead.

RE: Done it. Thank you.

Thank you for your final approval and acceptance